Andrzej S Tarnawski  
Editor-in-Chief  
*World Journal of Gastroenterology*

Dear Editor:

We wish to re-submit the manuscript titled “The efficacy of endoscopic ultrasound in the evaluation of small gastrointestinal stromal tumors.” The manuscript ID is 77878.

We thank you and the reviewers for your thoughtful suggestions and insights. The manuscript has benefited from these insightful suggestions. I look forward to working with you and the reviewers to move this manuscript closer to publication in *World Journal of Gastroenterology*.

The manuscript has been rechecked and the necessary changes have been made in accordance with the reviewers’ suggestions. The responses to all comments have been prepared and attached herewith/given below.

Reviewer1: Aside from small typographic errors (for example "illustrats" in the results section, first paragraph), this was excellently written.  
Answer: Thanks for your suggestions. We have revised these errors in the manuscript.

Reviewer2: Minor comments: 1. The manuscript requires a minor editing. Some small typographic errors should be corrected. 2. The aim of the study in the abstract should be written single.  
Answer: We have made some typographic corrections in the manuscript and written the aim of the study separately in the abstract.

Reviewer3: The reviewer recommends to accept this manuscript after a minor editing.  
Answer: Thanks for your suggestions. We have polished the manuscript and made a editing.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,  
Jintao Guo, M.D., Ph.D., Professor  
Department of Gastroenterology  
Shengjing Hospital of China Medical University  
Shenyang 110004, Liaoning Province, China  
Email: guojt@sj-hospital.org